Author’s response to reviews

Title: Evaluating the effectiveness, efficiency, cost and value of contacting study authors in a systematic review: a case study and worked example.

Authors:

Chris Cooper (christopher.cooper@exeter.ac.uk; chingfordhall@me.com)

Juan Talens Bou (juan.talens-bou2017@uni.strath.ac.uk)

Jo Varley Campbell (J.Varley-Campbell@exeter.ac.uk)

Version: 2 Date: 28 Nov 2018

Author’s response to reviews:

these are attached